Author's response to reviews

Title: Student centered curricular elements are associated with a healthier educational environment and lower depressive symptoms in medical students.

Authors:

Eiad Abdelmohsen AlFaris (efarix2@yahoo.co.uk)
Naghma Naeem (naghma.naeem@gmail.com)
Farhana Irfan (masoodfarhana1969@gmail.com)
Riaz Qureshi Qureshi (riaz.qureshi9@gmail.com)
Cees van der Vleuten (c.vandervleuten@maastrichtuniversity.nl)

Version: 5
Date: 22 July 2014

Author's response to reviews: see over
Dear Editor,

As advised in your communication dated July 5, 2014, we are uploading the revised version of our manuscript titled “MS 1050445339106753-Student centered curricular elements are associated with a healthier education-edited”. The manuscript has undergone copyediting by a fluent English speaking colleague. It also conforms to the journal style.

Hope it now meets the requirements.

Best Regards.

Dr Naghma Naeem

Corresponding Author